Author’s response to reviews

Title: Chinese siblings with hereditary medullary thyroid carcinoma caused by RET mutation: implications for RET oncogene detection

Authors:

Qin Huang (260425551@qq.com)
Aihua Hu (964599425@qq.com)
Mingsheng Zhang (zms75@163.com)

Version: 1 Date: 03 Dec 2019

Author’s response to reviews:

Dear editor,

Thanks for your letter, we had revised our manuscript according to your comments:

1. In the "Consent for publication" section, we clarified that: "Written informed consent for publication of this case report was obtained from all the patients whose details are published here. A copy of the written consent is available for review by the Editor of this journal."

2. We had removed the track changes or highlighting in our manuscript.

Best regards,

Mingsheng Zhang